Author’s response to reviews

Title: Advancing Laboratory Medicine in Hospitals through Health Information Exchange: A Survey of Specialist Physicians in Canada

Authors:

Louis Raymond (louis.raymond@uqtr.ca)
Eric Maillet (eric.maillet@usherbrooke.ca)
Marie-Claude Trudel (marie-claude.trudel@hec.ca)
Josianne Marsan (josianne.marsan@sio.ulaval.ca)
Ana Ortiz de Guinea (ana.ortiz-de-guinea@hec.ca)
Guy Paré (guy.pare@hec.ca)

Version: 3 Date: 13 Feb 2020

Author’s response to reviews:

Editor comments #1:

Please replace the footnotes with endnotes. Endnotes should be designated within the text using a superscript lowercase letter and all notes (along with their corresponding letter) should be included in the Endnotes section. Please format this section in a paragraph rather than a list.

Authors' response:

Done.

Editor's comment #2:

Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

Authors' response:

Done.
Editor's comment #3:

At this stage, please upload your proofread manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text. Should you wish to respond to these revision requests, please include the information in the designated input box only.

Authors' response:

Done.